Author’s response to reviews

Title: Successful treatment of paraneoplastic hypercalcaemia in a patient with giant condyloma acuminatum: a case report

Authors:

Thomas Linnemann (Thomas.Linnemann@klinikumbielefeld.de)
Frauke Müller (Frauke.Mueller@klinikumbielefeld.de)
Mathias Löhner (Mathias.Loehnert@klinikumbielefeld.de)
Peter Hirnle (Peter.Hirnle@klinikumbielefeld.de)
Martin Görner (Martin.Goerner@klinikumbielefeld.de)

Version: 5 Date: 25 July 2013

Author’s response to reviews:

Dear reviewers,

we made a point by point summary, where and how we have revised your manuscript. Thank you for your patience.